Author's response to reviews

Title: Glucocorticoid receptor mRNA and protein isoform alterations in the orbitofrontal cortex in schizophrenia and bipolar disorder

Authors:

Duncan Sinclair (d.sinclair@neura.edu.au)
Maree J Webster (websterm@stanleyresearch.org)
Janice M Fullerton (j.fullerton@neura.edu.au)
Cynthia Shannon Weickert (c.weickert@neura.edu.au)

Version: 2 Date: 20 February 2012

Author's response to reviews:

Since initially submitting the manuscript, the following change has been made:
Two sentences were added to the methods section: ‘These studies were carried out in accordance with the declaration of Helsinki, after approval by the Human Research Ethics Committee at the University of NSW (#HREC07261). Written informed consent for use of tissues in the study was obtained from next of kin’.